Author's response to reviews

Title: Socioeconomic status and the incidence of non-central nervous system childhood embryonal tumours in Brazil

Authors:

Beatriz de Camargo (bdecamar@terra.com.br)
Juliana MO Ferreira (ju.jmoferreira@gmail.com)
Rejane S Reis (rejane@inca.gov.br)
Sima Ferman (sferman@uol.com.br)
Marceli de Oliveira Santos (msantos@inca.gov.br)
Maria S Pombo-de-Oliveira (mpombo@inca.gov.br)

Version: 2 Date: 4 September 2010

Author's response to reviews: see over
Dear Editor,

Brazil is the largest country in Latin America. Until now, published information on childhood cancer incidence has been rather scanty, including information regarding embryonal cancer. Therefore, an analysis testing the correlation of cancer incidence in children from population-based cancer registries spread across most regions of the country and socioeconomic status would be of very great interest and importance for epidemiologists and researchers.

We hope to have the opportunity to present our data in BBC Cancer.

We had not submitted our data to any other journal.

Sincerely yours,

Beatriz de Camargo, Md, PhD
Programa de Hematologia-Oncologia Pediátricos, Instituto Nacional de Câncer
Rua André Cavalcanti, 37 Cep: 20231-050 Rio de Janeiro, RJ- Brasil
Tel: (21) 3233-1324 Fax: (21) 3233-1470
e-mail: bdecamar@terra.com.br